Page1 CS248 W98 Lecture 1Copyright � Pat HanrahanMotivationInteractive Graphical User InterfacesnDesktop metaphorInput: Mouse, keyboardOutput: CRTnImmersive interfaces

Ouput: HMD, RWB Pat HanrahanImmersive Interfaces

Page2

CS248 W98 Lecture 1Copyright � Pat HanrahanMotivationDigital Media TechnologiesTraditional media (analog to digital transition)nPrinting (done)nTelevision and video (underway) n Cameras and Film (underway) n Movies?

New media n Multimedia personal computer n

CS248 W98 Lecture 1Copyright � Pat HanrahanMotivationVisualizationExamplesnScience: Thunderstorm simulationnMathematics: Sphere eversion n Engineering: Computation fluid dynamics n Computer Science: Software system

Page3

CS248 W98 Lecture 1Copyright � Pat HanrahanVisualization: Thunderstorm Simulation Pat HanrahanVisualization: Outside In

Page4CS248 W98 Lecture 1Copyright � Pat HanrahanVisualization: NASA�s FAST Pat HanrahanVisualization: Visible Human

Page5CS248 W98 Lecture 1Copyright � Pat HanrahanVisualization: SeeSoft old -> blue

CS248 W98 Lecture 1Copyright � Pat HanrahanMotivationArt and Designn�Electronic Art�nDrawings, Paintings, Blueprintsn

Mechanical CAD

Entertainment n Movies: Jurassic Park

and Toy Story

Games and CDROM Titles

Visual Simulation and Training n Airplane, ship, automobile simulators

Page6CS248 W98 Lecture 1Copyright � Pat HanrahanInterdisciplinary: Theory and PracticeSciencenPhysics of light, color and appearancenProjective geometry and perspectiven

Engineering n Hardware: Graphics and media processors, displays n Software: Graphics libraries, window systems

Art and Perception n Color: harmony and perception n Composition, lighting, ...

CS248 W98 Lecture 1Copyright � Pat Hanrahan

Shading: reflection and lighting

Transformation: viewing

Hidden Surface Elimination

Animation: time

GeometryPipeline

Page7CS248 W98 Lecture 1Copyright � Pat HanrahanImaging

Texture Mapping

Image Composition

Intensity and Color Quantization

Framebuffer/Display

Rasterization and Sampling

Imaging

Pat Hanrahan

Graphics Software Architecture

GLUT App

Hardware

GLGLUXlibXGLXXm/XtGLXServerClientSee Programming OpenGL for the X Window System, M. Kilgard